Author's response to reviews

Title: Recognition of delirium in ICU patients A diagnostic study of the NEECHAM confusion scale in ICU patients

Authors:

Henny Immers (Henny.immers@hetnet.nl)
Marieke J Schuurmans (m.j.schuurmans@med.uu.nl)
Jaap J Bijl van der (j.j.vanderbijl@med.uu.nl)

Version: 4 Date: 25 November 2005

Author's response to reviews:

- Abstract is included in the manuscript
- Author list is included in the manuscript
- Tables are included in the manuscript
- Discussion and conclusion section is separated in discussion section and conclusion section